Author's response to reviews

Title: Effects of a community-based healthy heart program on increasing healthy women's physical activity: a randomized controlled trial guided by Community-based Participatory Research (CBPR)

Authors:

Raha Pazoki (pazkipg@bpums.ac.ir)
Iraj Nabipour (nabipourpg@bpums.ac.ir)
Nasrin Seyednezami (Nasrin_naz2001@yahoo.com)
Seyed Reza Imami (inabim@yahoo.com)

Version: 5 Date: 4 August 2007

Author's response to reviews:

Dear Sir,

Thank you very much for comments of formatting of our manuscript.

Regards

I. Nabipour, MD